Author's response to reviews

**Title:** Economic hardships in adulthood and mental health in Sweden. The Swedish National Public Health Survey 2009.

**Authors:**

Johanna Ahnquist (johanna.ahnquist@ki.se)
Sarah P Wamala (sarah.wamala@ki.se)

**Version:** 4 **Date:** 29 September 2011

**Author's response to reviews:** see over
Dear BMC Series Editorial Production Team,

We are very pleased to hear that our paper is now conditionally accepted for publication in your Journal. We have made the following changes to the format of our manuscript so that the published article conforms to the journal style:

- Added a sentence indicating that all authors have read and approved the final manuscript in the section describing author’s contributions.
- Renumbered the tables in ascending numerical order.
- Removed all highlighting and tracking from the manuscript.
- Changed the title “Introduction” to “Background”.
- Placed all tables in a section after the references.
- Changed the colour of the table lines to black.
- Re-layout of tables with merged elements.

We have also checked the manuscript for typographical errors.

Hopefully you will find the alterations we have made satisfactory. Please let us know if we need to make any further alterations.

Sincerely,

Johanna Ahnquist (BSc, MPH, PhD-student)
Karolinska Institutet and Swedish National Institute of Public Health
SE-831 40 Östersund
Sweden
Phone:+46 -63-19 96 91
E-mail address: johanna.ahnquist@ki.se